Author's response to reviews

Title: Effect of Changes in Contractility on the Index of Myocardial Performance in the Dysfunctional Left Ventricle

Authors:
  Steven J Lavine (steven.lvine@jax.ufl.edu)

Version: 2 Date: 6 November 2006

Author's response to reviews: see over
Dear Dr. Picano,

Enclosed is the revised manuscript entitled: “Effect of changes in contractility on the index of myocardial performance in the dysfunctional left ventricle with preserved left ventricular systolic function” for consideration for publication in Cardiovascular Ultrasound. I have addressed all the technical requests. With regard to the question posed regarding more severe LV dysfunction, this was further discussed at the end the “Limitations” section of the Discussion. I hope you find the all the changes made acceptable.

Thank you for your consideration.

Sincerely yours,

Steven J. Lavine, M.D.
Director, Noninvasive Laboratories
Cardiovascular Center
Associate Professor of Medicine
University of Florida/Jacksonville